# PROTEOMICS STUDY ON BIOMARKERS FOR HETEROTOPIC OSSIFICATION SECONDARY TO TRAUMATIC BRAIN INJURIES

Min HONGWEI, MD<sup>1,2</sup>, Qu TIEBING, MD<sup>1,2</sup>, Li ZHIGUO, MD<sup>3\*</sup> and Liu KEMIN, MD<sup>1,2\*</sup>

From the <sup>1</sup>School of Rehabilitation, Capital Medical University, <sup>2</sup>Department of Orthopedics and Rehabilitation, Beijing Bo'ai Hospital, China Rehabilitation Research Center, Beijing and <sup>3</sup>Department of Medical Research Center, International Science and Technology Cooperation Base of Geriatric Medicine, North China University of Science and Technology, Tangshan, China \*These authors contributed equally to this publication.

*Objective:* To identify differentially expressed serum proteins that could serve as sensitive biomarkers of heterotopic ossification in patients with traumatic brain injury.

Methods: From August 2014 to December 2015, 18 patients with traumatic brain injury were enrolled in the study, and blood samples were collected. Patients with traumatic brain injury were divided based on the presence (n = 9 patients, heterotopic ossification group) or absence (n = 9 patients, traumatic brain injury group or control group) of heterotopic ossification. Protein expression profiles were compared using 2-dimensional electrophoresis. Differentially expressed proteins were examined using matrix-assisted laser desorption/ionization and time-of-flight tandem mass spectrometry (MALDI-TOF/TOF). The differentially expressed proteins identified were further confirmed by Western blotting.

**Results:** Seven protein spots were differentially expressed between heterotopic ossification and traumatic brain injury groups in 2-dimensional electrophoresis analysis. Vitamin D binding protein (Gc protein), retinol binding protein 4 (RBP4) and haptoglobin expression decreased significantly in the heterotopic ossification group compared with the control group (p < 0.05), and this was further confirmed by Western blotting.

*Conclusion:* Lower levels of expression of Gc protein, RBP4 and haptoglobin may be closely related to heterotopic ossification after traumatic brain injury. These proteins may be potential biomarkers of heterotopic ossification secondary to traumatic brain injury.

*Key words:* traumatic brain injury; heterotopic ossification; proteomics, Gc protein, RBP4, haptoglobin.

Accepted Oct 24, 2019; Epub ahead of print Oct 29, 2019

J Rehabil Med 2020; 52: jrm00001

Correspondence address: Liu Kemin, Department of Orthopedics and Rehabilitation, Beijing Bo'ai Hospital, China Rehabilitation Research Center, Beijing, China. E-mail: keminlqliu@sina.com; and Li Zhiguo, International Science and Technology Cooperation Base of Geriatric Medicine, North China University of Science and Technology, Tangshan, China. E-mail: lizhigao@ncst.edu.cn

Traumatic brain injury (TBI) is common and its incidence is increasing. Clinical data have shown that 8–20% of TBI is complicated by heterotopic os-

# LAY ABSTRACT

Heterotopic ossification secondary to traumatic brain injury results in limited range of motion in the affected joint. There is no optimal treatment for controlling heterotopic ossification. It is therefore of great importance to diagnose this condition. In order to identify potential biomarkers for heterotopic ossification, this study used proteomic techniques to compare the plasma profile of patients with traumatic brain injury with and without heterotopic ossification. Expression levels of 3 proteins were significantly lower in the heterotopic ossification group than in the control group. These proteins may be potential biomarkers of heterotopic ossification.

sification (HO) (1). HO is defined as bone tissue that forms in the soft tissue around a joint. The exact pathophysiology of TBI-HO is not clear. However, several studies, such as those by Bidner et al. (2) and Kurer et al. (3), provide clear evidence for the hypothesis of a humoral mechanism for accelerated fracture healing with increased bone formation after TBI. Gautschi et al. (4) suggested an osteo-inductive effect of the cerebrospinal fluid from patients with severe TBI.

The onset of HO is accompanied by limited range of motion in the affected joint, and there is no optimal treatment for controlling the disease (5). The disability incurred as a result of HO is guite variable, and >20%of patients develop overt dysfunction as a result of soft-tissue loss, joint contractures, and chronic pain (6). Therefore, it is of great importance to diagnose this disease, and to find markers to determine whether HO is present after a TBI. At present, diagnosis of HO includes an analysis of a few biomarkers, such as serum alkaline phosphatase (7) and 24-h urinary prostaglandin levels (8). Creatine phosphokinase, prostaglandin E and matrix metalloproteinase 9 are potential biomarkers for neurogenic HO. However, the specificity and sensitivity of this analysis are poor (9–10). Proteomics has the potential to identify novel biomarkers from pathological tissues and biological fluids (11). Evidence exists that many disease processes are associated with specific quantitative and functional changes in proteins in pathological tissues (12). To find an easier and simpler diagnostic method and to identify pathogenic proteins associated with HO, we assessed plasma tissues from patients with TBI-HO for proteomic alterations.

### p. 2 of 7 M. Hongwei et al.

The pathogenesis of HO in the context of a TBI is currently unclear. The basic assumption is that stimulating factors induce mesenchymal cells to differentiate into osteoblasts (1, 13). Chalmers et al. proposed 3 conditions necessary for the formation of HO: an osteogenic inducer, a precursor cell of bone formation, and an environment that permits osteogenesis. HO is thought to arise from an imbalance in the promotion and inhibition of osteogenesis induced by local and multiple systemic stimulating factors (14). Zhuang et al. (15) found that low expression of the alpha-2-HS-glycoprotein B chain was closely related to HO formation after a hip replacement. They concluded that AHSG B chain might serve as a sensitive protein biomarker of HO. Eun et al. (16) found that 6 proteins, including PRO2675, were highly expressed in patients with ossification of the posterior longitudinal ligament. They suggested that PRO2675 might serve as a sensitive protein biomarker for ossification of the posterior longitudinal ligament. Rodenberg et al. (10) suggested that matrix metalloproteinase-9 was a biomarker for early extracellular recombination, and that it might be useful for the early diagnosis of HO formation.

This study used proteomic technics to compare the plasma profile of patients with TBI with and without HO, with the aim of identifying potential biomarkers for HO.

# **MATERIAL AND METHODS**

#### Patients

From August 2014 to December 2015, a total of 18 patients with TBI were enrolled in the study. Of these 18 patients, 9 had HO (HO group) and 9 did not have HO (TBI group). Plasma samples were collected from each patient. Inclusion criteria for the TBI group were: age range 18–60 years; diagnosis of TBI according to clinical presentation and imaging findings (computed tomography (CT) and magnetic resonance imaging (MRI)); no serious hepatic, renal, hematopoietic, endocrine or psychiatric diseases; when coma was present, Glasgow Coma Score was between 3 and 12; and no HO was present before the TBI occurred. Inclusion criteria for the TBI group were, in the early stage (4–12 months after TBI), no signs of infection, disease progression and reduced range of motion in the affected joint; and X-rays of affected joints revealed HO.

Exclusion criteria were: any fractured limbs; any trauma (including burns) or previous surgery in the affected joints; accompanying spinal cord injury; or accompanying intracranial infection, myelitis, haemophilia, ankylosing spondylitis, senile idiopathic hyperostosis, diffuse idiopathic skeletal hyperostosis, ossification of the posterior longitudinal ligament, ossification of the ligamentum flavum or soft tissue ossification diseases.

Informed consent was obtained from all participants before collecting blood samples. The study was approved by the ethics committee of the China Rehabilitation Research Center (number 2015k056).

#### Plasma sample

Venous blood samples (5 ml) from each patient were placed in anticoagulant K2EDTA tubes. Within 2 h each sample was centrifuged at 4°C at 1,300 relative centrifugal force (RCF) for 10 min to remove blood cells. The supernatant was centrifuged at 2,400 RCF for 15 min to remove potentially remaining cellular material and platelets, in order to prepare platelet-poor plasma. The plasma was aliquoted and stored at  $-80^{\circ}$ C until use.

#### Two-dimensional electrophoresis for protein separation

In order to reduce proteomic study heterogeneity and experimental cost, every 3 plasma samples were mixed randomly to form a new sample. Thus, a total of 3 mixed samples were prepared for each group (HO and TBI groups, 9 cases each). Each plasma sample was separated individually by 2-dimensional electrophoresis and the differentially expressed proteins were identified by MALDI-TOF/TOF, as described previously (17). Briefly, 1.3 mg each of experimental group and control group plasma protein was loaded onto an immobilized pH gradient gel strip (pH3-10NL, 24 cm; GE Healthcare Life Sciences, Pittsburgh, PA, USA) using the in-gel rehydration mode. Protein was separated by first-dimensional electrophoresis (Ettan IPGphor 3 Isoelectric Focusing Unit, GE Healthcare Life Sciences), followed by 13%T and 3%C second-dimensional separation (Ettan DALT II System, GE Healthcare Life Sciences). Separated protein spots were visualized using colloidal Coomassie blue staining according to the manufacturer's protocol. Images were analysed using Image Master 2D Platinum 6.0 software and protocol (GE Healthcare Life Sciences). To identify differentially expressed proteins, the differentially expressed protein spots were cut manually from the gels. Each excised protein spot was digested in-gel and identified by MALDI-TOF/TOF follow MASCOT analysis.

## Western blot analysis

To avoid the uncertainties caused by sample mixture, all plasma samples were analysed by Western blot assay. Briefly, the protein sample (15 µg) was separated using a 12% polyacrylamide gel and electro-transferred to a nitrocellulose membrane in a Trans-Blot Semi-Dry Electrophoretic Transfer Cell (Bio-Rad, Hercules, CA, USA). Non-specific bands were blocked in TBS-T (25 mM Tris; 150 mM NaCl; 0.05% Tween20, pH 7.5) containing 5% skimmed milk at room temperature for 1 h. Membranes were subsequently incubated with the primary antibody anti-GC (1:1000 dilution; ABclonal), anti-HP (1:1000 dilution; Abcam), and anti-retinol binding-protein 4 (RBP4) (1:1000 dilution; ABclonal), and  $\beta$ -actin (1:5000 dilution; Sungene Biotech) overnight at 4°C, followed by incubation with anti-mouse or anti-rabbit horseradish peroxidise conjugated IgG at room temperature for 1 h. The immunocomplexes were visualized by chemiluminescence using the Amerisham ECL Western Blotting Detection kit (GE Healthcare Life Sciences). The signals were acquired by the molecular imager ChemiDocXRS+ (Bio-Rad) and then quantified using Quantity-One software, version 4.31 (Bio-Rad).

#### Statistical analysis

Statistical analysis was carried out using Student's *t*-tests and false discovery rate (FDR), as described by Biron et al. (2). Data were expressed as the mean (standard deviation; SD), and all analyses were performed with SPSS 17. Groups were compared using a *t*-test for 2 independent samples. *p*-values <0.05 were considered statistically significant.

Biomarkers for heterotopic ossification secondary to TBI p. 3 of 7

isease, months Glasgow Coma Score
Mean (SD)
9) 6.67 (1.58)
) 6.29 (1.80)
0.45
0.659
_ 9 )

Table I. Comparison of characteristics between patients with heterotopic ossification (HO group) and control group

 $\chi^2$  and *p*-values for each characteristic or ratio (i.e. male/female, yes/no) indicate the significance of differences between the TBI-HO and TBI groups. TBI: traumatic brain injury; SD: standard deviation.

# RESULTS

The HO group comprised 2 males and 7 females, with a mean age of 52 years (SD 10.2). The mean disease course was 6.3 months (SD 0.5). The control group comprised 3 males and 6 females, with a mean age of 50 years (SD 9.2). Analysis of the baseline data of the 2 groups showed no significant differences in sex, age, course of the disease, or disease severity for HO (Table I).

Representative 2-dimensional gel electrophoresis (2-DE) patterns of plasma are shown in Fig. 1A. In the maps, most of the proteins are distributed in an area of pI 4.0–8.0 and molecular weight 14–97 kDa (Fig.

1A). Seven protein spots were found to be differentially expressed between the 2 groups (Fig. 1B). These spots were submitted for MALDI-TOF/TOF assay. Five proteins, with their MASCOT search results supporting identification data, are listed in Table II. The Mascot scores of the 5 proteins are >60, which indicates that they are less likely to be wrongly identified (p<0.05). In addition, compared with calculated pI and molecular weight, their pI variations is <2 and molecular mass variation is less than 30%. These results imply that the possibility of incorrect identification of these proteins is very small.



Fig. 1. Differentially expressed plasma proteins between traumatic brain injury (TBI) and heterotopic ossification (HO) groups. (A) Comparison of plasma 2-DE patterns between TBI with and without HO. (B) Magnified comparison map of differentially expressed proteins.

#### p. 4 of 7 M. Hongwei et al.

#### Table II. Results of MALDI-TOF/TOF analysis of the differentially expressed proteins

Spot number <sup>a</sup>	Accession number	Protein name	Gene name	Score <sup>b</sup>	pI O <sup>c</sup>	MWO <sup>d</sup>	pI <sup>c,e</sup>	MW <sup>c,f</sup>	Matches <sup>g</sup>	Adj. <i>p</i> -value (FDR) <sup>h</sup>	Fold ratio <sup>I</sup>
1	P02774.1	Vitamin D binding protein	GC	633	5.3	68	5.4	53	11 (6)	0.030	0.60
2					5.0	47				0.034	0.56
3	P00738	Haptoglobin	HP	284	5.6	51	6.1	45	6 (4)	0.029	0.56
4					4.9	48				0.027	0.72
5	P02753	Retinol binding protein 4	RBP4	262	5.4	23	5.7	23	5 (3)	0.041	0.65
6	P0DJI8	Serum amyloid A-1 proteir	n SAA1	464	5.7	14	6.3	14	7 (3)	0.026	3.72
7	P0DJI8	Serum amyloid A-1 proteir	n SAA1	377	6.0	14	6.3	14	5 (4)	0.039	7.81

<sup>a</sup>Spot numbers were defined according to spot positions in 2-dimensional (2-D) electrophoresis gels.

<sup>b</sup>Score for the identified protein, calculated as -10\*Log (P), where P is the probability that the observed match is a random event.

<sup>c</sup>Observed protein isoelectric point in 2-D gels. <sup>d</sup>Observed protein molecular weight in 2-D gels.

<sup>e</sup>Calculated protein isoelectric point.

<sup>f</sup>Observed protein molecular weight.

<sup>9</sup>Number of total mass values searched (number of matched mass values).

<sup>h</sup>Adjust *p*-value was calculated using Student's *t*-tests and false discovery rate (Benjamini–Hochberg)

Fold ratios are shown as a protein expression ratio of heterotopic ossification/traumatic brain injury.

MALDI-TOF/TOF: matrix-assisted laser desorption/ionization and time-of-flight tandem mass spectrometry.

The differential expression of GC, HP and RBP4 were further validated by Western blot analysis. As shown in Fig. 2, the expression levels of GC, HP and RBP4 were similar to those observed in the 2-DE gels. The abundance of protein was significantly altered (p < 0.05) in the plasma of patients with HO.

# DISCUSSION

Protein separation using 2-dimensional electrophoresis in this study showed that GC, HP, and RBP4 are differentially expressed in the plasma of patients with HO. These results were confirmed by Western blot analysis. The findings may contribute to improved understanding of the patho-



between traumatic brain injury (TBI) and heterotopic ossification (HO) groups. Abundances were quantified for (top left) vitamin D binding protein (GC), (top right) retinol binding protein 4 (RBP4), and (bottom left) haptoglobin (HP). Top panels: Western blots show plasma samples probed with specific antibodies against the indicated proteins. Bottom panels: comparisons of mean±standard deviation (SD) volumes in patients with TBI-HO and with TBI alone. \*p < 0.05.

Journal of Rehabilitation Medicine

GC

JRM

genesis of TBI-HO. Moreover, they may represent new biomarkers for the diagnosis of HO secondary to TBI.

In 1959, a new plasma protein was first discovered by Hirschfeld, using immunological methods. This protein was named the "Gc protein" according to its immunological properties. It is also known as "vitamin D binding protein", and its concentration in plasma is high. Gc protein is mostly secreted by liver cells and it possesses a number of biochemical functions. It facilitates binding between actin and vitamin D and it strengthens the chemotactic activity of C5 for neutrophils (18). A number of studies have shown that polymorphisms in the vitamin D receptor gene may be associated with ossification of the posterior longitudinal ligament (19, 20). Moreover, Gc protein-derived macrophage activating factor was shown to stimulate osteoclast activity and promote bone resorption (21, 22). Those activities suggest that the Gc protein might influence the formation of HO by affecting osteoclast activity.

In 2005, Yang et al. (23) reported the identification of a new adipose tissue factor, known as RBP4, RBP4 is a single peptide chain, mainly generated by fat cells in liver and placenta. The function of RBP4 is to transport small-molecular-weight hydrophobic proteins. It is also involved in vitamin A metabolism and storage. Interestingly, a phenomenon called "shifting" has been observed in many human diseases, where bone tissue is replaced with adipose tissue. For example, in ageing, long-term use of corticosteroids and long-term immobilization, reductions in bone formation are accompanied by increases in bone marrow adipose tissue. Both osteoblasts and adipose cells are derived from bone marrow mesenchymal stem cells. The shifting phenomenon suggests that the differentiation of bone marrow mesenchymal stem cells into adipogenic and osteogenic lineages are not isolated processes; instead, there appears to be a balance between these processes (24). Studies of adipocytes have shown that inhibiting RBP4 expression promoted adipogenic differentiation (25). Conversely, retinol could induce osteogenic differentiation and inhibit adipogenic differentiation in mesenchymal stem cells (26). These data suggest that RBP4 may play a role in osteogenic differentiation. On the other hand, RBP4 may be involved in the balance between bone resorption and bone formation by increasing bone resorption. The underlying mechanism may involve direct vitamin A stimulation of osteoclast formation (27). Because the vitamin A receptor and the 1,25 dihydroxy vitamin D3 receptor belong to the steroid and thyroid hormone receptor family, their functions interfere with each other. Vitamin A may have an indirect negative effect on skeletal tissue, by regulating calcium levels and bone mineralization in ways that interfere with the normal function of vitamin D (28). Moreover, the correlation between RBP4 and the osteoclast index may be related to increases in inflammatory factors. For example, a variety of factors can stimulate the secretion of cytokines, such as interleukin 6 (IL-6) and tumour necrosis factor alpha (TNF- $\alpha$ ) (9), which enhance the differentiation, maturation and activity of osteoclasts and increase the levels of free fatty acids. These conditions can promote bone resorption (30) and increase bone turnover, which then lead to an imbalance in bone formation and bone resorption processes.

Haptoglobin (Hp) is a highly abundant plasma glycoprotein. Its main function is to combine with free haemoglobin (Hb) to form stable Hp-Hb complexes. Mediated by the scavenger receptor CD163, these complexes are rapidly degraded by mononuclear phagocytes to protect DNA and tissues from oxidative damage (31, 32). Hp also regulates angiogenesis and arterial remodelling (33). These activities may contribute to HO, because revascularization is important to HO formation (34). In addition, a bioinformatics analysis of Hp showed that Hp is important in the molecular regulation of inflammation (35, 36). This finding suggests that Hp may play an important role in the inflammatory process of fracture healing, which is similar to the process of HO formation. A series of studies by Stevens et al. (37) showed that IL-1 and TNF- $\alpha$  could indirectly promote the expression and activity of Hp by stimulating IL-6 synthesis. In turn, Hp could inhibit the activities of cyclooxygenase and lipoxygenase, and thus inhibit the synthesis of inflammatory factors, such as leukotrienes and prostacyclin (38). Recent studies have found that prostaglandin E2 could prevent osteoblast differentiation (39). Presumably, Hp might interfere with the formation of HO through this pathway. In addition. Hp prevented the destruction of hyaluronic acid (HA), mediated by oxygen-derived free radicals, by forming HA-Hp complexes (40). This activity may also be linked to HO formation, as it has been reported that free radical scavengers could prevent formation of HO (41).

# Limitation

This study has some limitations. First, it was designed as a case-control study; therefore, the results can only show a positive correlation between TBI-HO and low expression levels of Gc, HP and RBP4; they cannot indicate whether a causal relationship exists. Prospective studies are needed to clarify the relationship between TBI-HO and these factors. Secondly, the small size of this study might have limited the detection of some differences between groups. In addition, we did not simultaneously monitor the body's vitamin status

#### p. 6 of 7 M. Hongwei et al.

or the levels of inflammatory factors associated with bone metabolism; thus, many confounding factors may have affected the results or contributed to the lack of convincing results in this study.

This study showed that 3 proteins were expressed at a low level in the HO group. Because of the limitations of a case-control study, it is not possible to infer that low expression of protein is the origin or result of HO. This issue requires further in-depth research through prospective cohort study. If these biomarkers can diagnose HO early, then corresponding preventive measures could be taken to reduce or delay the occurrence of HO, reduce complications and disabilities, and improve quality of life in patients with TBI.

# Conclusion

This study is the first to discover biomarkers for TBI-HO through the use of comparative proteomics. These results may contribute to increased understanding of the pathogenesis of TBI-HO. Moreover, these findings suggest that Gc, HP and RBP4 may be potential biomarkers for HO.

# REFERENCES

- Cipriano CA, Pill SG, Keenan MA. Heterotopic ossification following traumatic brain injury and spinal cord injury. J Am Acad Orthop Surg 2009; 17: 689–697.
- Bidner SM, Rubins IM, Desjardins JV, Zukor DJ, Goltzman D. Evidence for a humoral mechanism for enhanced osteogenesis after head injury. J Bone Joint Surg Am 1990; 72: 1144–1149.
- Kurer MH, Khoker MA, Dandona P. Human osteoblast stimulation by sera from paraplegic patients with heterotopic ossification. Paraplegia 1992; 30: 165–168.
- Gautschi OP, Toffoli AM, Joesbury KA, Skirving AP, Filgueira L, Zellweger R. Osteoinductive effect of cerebrospinal fluid from brain-injured patients. J Neurotrauma 2007; 24: 154–162.
- 5. Meyers C, Lisiecki J, Miller S, Levin A, Fayad L, Ding C, et al. Heterotopic ossification: a comprehensive review. JBMR Plus 2019; 27: e10172.
- Balboni TA, Gobezie R, Mamon HJ. Heterotopic ossification: pathophysiology, clinical features, and the role of radiotherapy for prophylaxis. Int J Radiat Oncol Biol Phys 2006; 65: 1289–1299.
- Gautschi OP, Cadosch D, Frey SP, Skirving AP, Filgueira L, Zellweger R. Serum-mediated osteogenic effect in traumatic brain-injured patients. ANZ J Surg 2009; 79: 449–455.
- Schurch B, Capaul M, Vallotton MB, Rossier AB. Prostaglandin E2 measurements: their value in the early diagnosis of heterotopic ossification in spinal cord injury patients. Arch Phys Med Rehabil 1997; 78: 687–691.
- Garland DE. A clinical perspective on common forms of acquired heterotopic ossification. Clin Orthop Relat Res 1991; 263: 13–29.
- Rodenberg E, Azhdarinia A, Lazard ZW, Hall M, Kwon SK, Wilganowski N, et al. Matrix metalloproteinase-9 is a diagnostic marker of heterotopic ossification in a murine model. Tissue Eng Part A 2011; 17: 2487–2496.
- Hanash SM, Bobek MP, Rickman DS, Williams T, Rouillard JM, Kuick R, et al. Integrating cancer genomics and proteomics in the post-genomeera. Proteomics 2002; 2: 69–75.
- 12. Murphy S, Zweyer M, Mundegar RR, Swandulla D, Ohlen-

dieck K. Proteomic serum biomarkers for neuromuscular diseases. Expert Rev Proteomics 2018; 15: 277–291.

- Toffoli AM, Gautschi OP, Frey SP, Filgueira L, Zellweger R. From brain to bone: evidence for the release of osteogenic humoral factors after traumatic brain injury. Brain Inj 2008, 22: 511–518.
- Chalmers J, Gray DH, Rush J. Observations on the induction of bone in soft tissues. J Bone Joint Surg Br 1975; 57: 36–45.
- Zhuang Y, Zhang X, Xu W, Lin S, Qiu M, Lin Z. Proteomics on heterotopic ossification after hip arthroplasty. J Trauma Surg 2014, 16: 147–150.
- Eun JP, Ma TZ, Lee WJ, Kim MG, Yoo MJ, Koh EJ, et al. Comparative analysis of serum proteomes to discover biomarkers for ossification of the posterior longitudinal ligament. Spine (Phila Pa 1976) 2007; 32: 728–734.
- 17. Zychowicz ME. Pathophysiology of heterotopic ossification. Orthop Nurs 2013; 32: 173–178.
- Zeiser JJ, Klodmann J, Braun HP, Gerhard R, Just I, Pich A. Effects of Clostridium difficile Toxin A on the proteome of colonocytes studied by differential 2D electrophoresis. J Proteomics 2011; 75: 469–479.
- Shiigi E, Sugiyama T, Tanaka H, Murata H, Shirakura Y, Kawai S. Possible involvement of vitamin D receptor gene polymorphism in male patients with ossification of spinal ligaments. J Bone Miner Metab 2001; 19: 308–311.
- 20. Kobashi G, Ohta K, Washio M, Okamoto K, Sasaki S, Yokoyama T, et al. FokI variant of vitamin D receptor gene and factors related to atherosclerosis associated with ossification of the posterior longitudinal ligament of the spine: a multi-hospital case-control study. Spine (Phila Pa 1976) 2008; 33: E553–E558.
- 21. Brown AJ, Zhang F, Cooke NE, Ritter CS. The role of the serum vitamin D binding protein in the actions of the vitamin D analog eldecalcitol (ED-71) on bone and mineral metabolism. Calcif Tissue Int 2013; 93: 163–171.
- Ikeda T, Kasai M, Tatsukawa E, Kamitakahara M, Shibata Y, Yokoi T, et al. A bone substitute with high affinity for vitamin D-binding protein-relationship with niche of osteoclasts. J Cell Mol Med 2014; 18: 170–180.
- Yang Q, Graham TE, Mody N, Preitner F, Peroni OD, Zabolotny JM, et al. Serum retinol binding protein 4 contributes to insulin resistance in obesity and type 2 diabetes. Nature 2005; 436: 356–362.
- Fan Q, Tang T, Zhang X, Dai K. The role of CCAAT/enhancer binding protein(C/EBP) alpha in osteogenesis of C3H10T1/2 cells induced by BMP- 2. J Cell Mol Med 2009; 13: 2489–2505.
- 25. Munkhtulga L, Nagashima S, Nakayama K, Utsumi N, Yanagisawa Y, Gotoh T, et al. Regulatory SNP in the RBP4 gene modified the expression in adipocytes and associated with BMI. Obesity (Silver Spring) 2010; 18: 1006–1014.
- Hisada K, Hata K, Ichida F, Matsubara T, Orimo H, Nakano T, et al. Retinoicacid regulates commitment of undifferentiated mesenchymal stem cells into osteoblasts and adipocytes. J Bone Miner Metab 2013; 31: 53–63.
- Scheven BA, Hamilton NJ. Retinoic acid and 1, 25-dihydroxyvitamin D3 stimulate osteoclast formation by different mechanisms. Bone 1990; 11: 53–59.
- 28. Evans RM. The steroid and thyroid hormone receptor supeffamily. Science 1988; 240: 889–895.
- Tzaphlidou M, Kafantari H. Influence of nutritional factors on bone collagen fibrils in ovariectomized rats. Bone 2000; 27: 635–638.
- Rosen CJ, Bouxsein ML. Mechanisms of disease: is osteoporosis the obesity of bone? Nat Clin Pract Rheumatol 2006; 2: 35–43.
- Wada T, Oara H, Watanabe K, Kinoshita H, Yachi A. Autoradiographic study on the site of uptake of the haptoglobin-hemoglobin complex. J Reticuloendothel Soc 1970; 8: 185–193.
- 32. Lim YK, Jenner A, Ali AB, Wang Y, Hsu SI, Chong SM, et al. Haptoglobin reduces renal oxidative DNA and tissue

Journal of Rehabilitation Medicine

JRM

damage during phenylhydrazine-induced hemolysis. Kidney Int 2000; 58: 1033–1044.

- Smeets MB, Pasterkamp G, Lim SK, Velema E, van Middelaar B, de Kleijn DP. Nitric oxide synthesis is involved in arterial haptoglobin expression after sustained flow changes. FEBS Lett 2002; 529: 221–224.
- Dilling CF, Wada AM, Lazard ZW, Salisbury EA, Gannon FH, Vadakkan TJ, et al. Vessel formation is induced prior to the appearance of cartilage in BMP-2-mediated heterotopic ossification. J Bone Miner Res 2010; 25: 1147–1156.
- Schindeler A, McDonald MM, Bokko P, Little DG. Bone remodeling during fracture repair: the cellular picture. Semin Cell Dev Biol 2008; 19: 459–466.
- Kushner I, Edgington TS, Trimble C, Liem HH, Muller-Eberhard U. Plasma hemopexin homeostasis during the acute phase response. J Lab Clin Med 1972; 80: 18–25.
- 37. Stevens AL, Wishnok JS, Chai DH, Grodzinsky AJ, Tannenbaum SR. A sodium dodecyl sulfate-polyacrylamide gel electrophoresis-liquid chromatography tandem mass spectrometry analysis of bovine cartilage tissue response

to mechanical compression injury and the inflammatory cytokines tumor necrosis factor alpha and interleukin-1beta. Arthritis Rheum 2008; 58: 489–500.

- Mastui SM, Greenberg HB (1996) Astroviruses. In: Fields BN, Knipe DM, Howley PM, Chanock RM, Melnick JL, Monath TP, Roizman B, Straus SE (eds) Fields virology. 3rd edn. Philadelphia; Lippincott-Raven Publishers: 2019. p. 811–824.
- Choudhary S, Blackwell K, Voznesensky O, Deb Roy A, Pilbeam C. Prostaglandin E2 acts via bone marrow macrophages to block PTH-stimulated osteoblast differentiation in vitro. Bone 2013; 56: 31–41.
- Chen SY, Chang YC, Lee YS, Chao HC, Tsao KC, Lin TY, et al. Molecular epidemiology and clinical manifestations of viral gastroenteritis in hospitalized pediatric patients in Northern Taiwan. J Clin Microbiol 2007; 45: 2054–2057.
- 41. Vanden Bossche L, Van Maele G, De Cock K, Vertriest S, Wojtowicz I, Vanderstraeten GG. Free radical scavengers have a preventive effect on heterotopic bone formation following manipulation of immobilized rabbit legs. Eur J Phys Rehabil Med 2008; 44: 423–428.